Author's response to reviews

Title: Development and psychometric properties of a scale for measuring internal participation from a patient and health care professional perspective

Authors:

Mirjam Koerner (mirjam.koerner@medsoz.uni-freiburg.de)
Markus Wirtz (markus.wirtz@ph-freiburg.de)

Version: 2 Date: 11 June 2013

Author's response to reviews:

Please consider that the title of the manuscript has been changed in the process of the revision.